CENWP-OP-B                                                                                                                                           25 May 2003

MEMORANDUM FOR CENWP-OP, ATTN: Gary Johnson

SUBJECT: Bonneville Lock and Dam, Fishway and Fish Activities for week 21 of 2003, which covers the period 18 through 24 May 2003.

1.  OPERATION SUMMARY.

1.1 Daily river flow averaged 226.6 to 272.4 kcfs.  Daily project tailwater elevation averaged 17.9’ to 20.3’ msl.  Daily project forebay elevation averaged 75.2’ to 75.9’ msl.  River temperature averaged 52 to 54° F.  Daily turbidity measured 4.0’ to 5.0’ on the secchi disk.

1.2 Spill Bays 1 and 18, which supply attraction flow for Cascade Island and Bradford Island fishways respectively, are included in the spill pattern.  Spill averaged 95.6 to 148.7 kcfs.

1.3 Unit Operation

1.3.1 Unit 1 remains out of service for turbine rehab.

1.3.2 Unit 13 was out of service from 1042 until 1330 on 19 May for gap loss testing.

1.3.3 Unit 14 went out of service at 0754 on 21 May due to the power supply for the exciter being moved to Unit 15.

1.3.4 Unit 15 returned to service at 0842 on 21 May.  Unit 15 was out of service from 0642 until 1624 on 22 May for gap loss testing equipment installation.

1.3.5 Unit 17 was out of service from 0830 until 1624 on 22 May for gap loss testing equipment removal.

1.4 Second Powerhouse fish units provide attraction flow for the Washington Shore fish ladder.  During fish passage season, the units are placed on standby from approximately 0000-0300 nightly if drawdown exceeds 1.0’.  

1.5 Gatewell drawdowns are measured on Wednesdays.

2.  ACTIVITIES. 
2.1 AWS Closures.  The following AWS valves were closed for cleaning this reporting week.

2.1.1 FV3-7 from 0845 until 0855 on 19 May.

2.1.2 FV3-9 from 1045 until 1113 on 19 May.

2.2 STS/VBS Inspections.  No STS/VBS inspections were conducted this reporting week.

2.3 Dewatering and Fish Salvage.  No dewatering or fish salvage activities this reporting week.

3.  RESEARCH.

3.1 CRITFC.  Sampling of chinook, sockeye, and coho continues until 31 October.

3.2 NMFS Lamprey.  Tracking continues once a month.  Trapping begins in June.

3.3 OCFWRU.  Fish collection continues. 

3.4 ODFW Steelhead.  Trapping to begin next week. 

3.5 PSMFC at DSM1.  Continuous PIT tag interrogation continues.  GBT and monitoring continue.

3.6 PSMFC at the SMF.  Monitoring personnel observed one kelt stuck on the separator bars.  107 steelhead, seven salmon, six unknown salmonids and 667 smolts slide over the separator bars.

3.7 University of Idaho.  Adult tracking tagging and tracking continue.  

3.8 U of I Transportation Study.  Evaluation of river run vs. barged Snake River juveniles to occur May-June.  

3.9 University of Washington.  Northern pikeminnow tagging and tracking from May through July.  Will hold fish in the FERL raceways.

3.10 USGS Lamprey.  Collection to begin in June.

3.11 WDFW jaw tagging.  Tagging steelhead and spring chinook for tooth tangle net studies.

3.12 WDFW Fish Counting.  16-hr visual counts at Washington Shore and Bradford Island.

3.13 WES/Battelle/BAE systems Juvenile hydroacoustic monitoring.  Gap loss monitoring continues. 

4.  FISHWAYS 
4.1 Project biologists inspected 18 through 24 May this week.  

4.2 Fish Passage Plan Observations.

	DATE
	LOCATion
	FPP Deviation
	CAUSE
	RESPONSE
	TR/WAR

	All week
	PH1 PCC
	Velocity <1.5fps
	Meter in slowest part of channel
	No response at this time.
	

	All week
	PH2 PCC
	No velocity measure.
	Meter is missing
	See misc. 5.1
	

	5/18
	BI fishways
	Reading <1.0’
	Unknown
	Operators to check.
	


4.3 Fish Ladder Temperatures.  Measurements are taken three weirs down from the count station at Bradford Island and at the junction of the UMT and the main ladder at Washington shore.  Temperatures are in Kelvin.  http://students.washington.edu/kyle/temptable.jpg 

	Date
	Wash. Shore
	Bradford Is.

	5/18
	285.2
	285.1

	5/19
	285.3
	285.3

	5/20
	285.6
	285.5

	5/21
	286.0
	286.0

	5/22
	286.2
	286.2

	5/23
	286.4
	286.4

	5/24
	286.5
	286.5


4.4 Sensor Calibration.  Calibration was checked on 22 May this reporting week.

4.5  Zebra Mussel Monitoring.  No signs of colonization were observed this reporting week. 

5. MISCELLANEOUS

5.1 Divers discovered the PH2 velocity meter was missing from the collection channel.  A new meter will be installed during the mid season dives this summer.

5.2 Sea lions and harbor seals have been observed in the tailrace of both powerhouses and the spillway.  No action is being taken by the project to harass, harm, or otherwise annoy these animals.  FFU is monitoring and recording their activity.

5.3 The SMF PDS screen cleaner is still out of service.  Currently waiting on a new engineering design.  Smolt monitoring personnel are cleaning the screen by hand until the screen cleaner is back in service.

JAMES R. MAHAR, P.E.

Operations Manager
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